



# Article The Impact of Market and Non-Market-Based Environmental Policy Instruments on Firms' Sustainable Technological Innovation: Evidence from Chinese Firms

Jie Jiang <sup>1</sup>,\*, Qihang Zhang <sup>1</sup> and Yifan Hui <sup>2</sup>

- <sup>1</sup> School of Intellectual Property, Nanjing University of Science and Technology, Nanjing 210014, China
- <sup>2</sup> School of Mathematics and Statistics, University of Glasgow, Glasgow G12 8QQ, UK
- \* Correspondence: eve.jie@outlook.com

**Abstract:** A firm's sustainable technological innovation (STI) is an important strategy to cope with the global challenges of the climate emergency and resource constraints. To encourage firms to pursue sustainable innovation, the government put its efforts into designing a proper environmental policy (EP). According to Porter's hypothesis, a well-designed and flexible EP will advance the pace of a firm's innovation. This paper argues that a flexible EP portfolio combining market and non-market-based EP instruments may affect a firm's STI. Market-based EP instruments are cost-effective and consistent from a long-term view, whereas non-market-based EP instruments are more forceful and effective in the short term. Hence, these two kinds of EP instruments could complement each other. Furthermore, technical executives in top management teams will moderate the relationships between EP instruments and firms' STI. Data analysis results of 618 Chinese public firms, who constantly participated in R&D activities during 2015–2019, supported these hypotheses. Contributions to EP and firm innovation theory, as well as suggestions for policymakers and firms' top management teams, are discussed.

Keywords: environmental policy; policy mix; sustainable technological innovation; technical executives

## 1. Introduction

High energy consumption and high emissions from industrialization pose an increasing threat to pollution and global warming. A firm's sustainable technological innovation (STI) is an important strategy to cope with the global challenges of pollution and resource constraints [1]. The challenge of approaching a firm's sustainability relies on the firm's STI [2,3]. Unlike technological innovation (TI), which merely focuses on upgrades and technological advances, STI refers to constantly gaining advantages through technological innovation and commercialization while respecting the environment [4,5]. This is in contrast to some research that argued that green patent numbers indicate STI performance. That is only a part of STI, which requires incessant R&D inputs and constantly profiting from the patentability and commercialization of the outputs while respecting the environment. STI falls under the umbrella of sustainable innovation (SI), which is a broader conception including making deliberate improvements to a firm's products, services, or business operations to produce long-term social and environmental advantages while generating financial success [6]. Thus, these long-term goals should be based on a proper EP framework that seeks to balance solving environmental issues and to boost technological innovation [7–9]. Previous research might discuss various utilized EP instruments and new tools [10,11]. However, the harmonization mechanism of non-market-based and market-based EP tools is underexplored. Moreover, abundant research has been expanded in discussing the relationship between EP instruments and SI from the viewpoint of sustainable green technological innovation (SGTI) and sustainable green business model innovation. Nevertheless, it is an undeveloped research question about how a flexible EP



Citation: Jiang, J.; Zhang, Q.; Hui, Y. The Impact of Market and Non-Market-Based Environmental Policy Instruments on Firms' Sustainable Technological Innovation: Evidence from Chinese Firms. *Sustainability* **2023**, *15*, 4425. https://doi.org/10.3390/su15054425

Academic Editor: Gaetano della Corte

Received: 27 January 2023 Revised: 23 February 2023 Accepted: 24 February 2023 Published: 1 March 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). portfolio affects the STI, specifically keeping a watchful eye on the balance of constant technological progress efficiency with sustainable green technology outputs.

Market-based and command-and-control policy instruments are the two most popular EP instruments to encourage firms' STI. Market-based policy instruments, such as environmental taxation and emission trading, are designed to fit the market mechanism. In contrast, command-and-control policy instruments are non-market-based policy instruments, such as environmental protection investment (EPI) and pollution fines. The OECD (2013) has been promoting the use of market-based instruments for many years, including taxes and emission trading systems. These instruments are regarded as the cheapest way for societies to reduce emissions. In 2019, EU environmental tax revenue amounted to EUR 330.6 billion, accounting for 5.9% of total government revenue from taxes and social contributions.

Similarly, the United States stopped its command-and-control regulatory approach in which the government requires individual plants to install specific control technologies to reduce pollution, regardless of varying costs of controls among the plants. Instead, in 1990, the United States passed the Clean Air Act Amendments and followed a market-based EP approach, known as cap-and-trade or "allowance trading", to address the acid rain problem [12]. According to these practices, countries prefer market-based EP instruments.

The Chinese government has pursued command-and-control regulation for several decades. The Chinese government has intensively used command-and-control regulation to prevent firms from overexploiting natural resources and maintain sustainable development. For example, firms' business costs include money invested in afforestation and penalty fees by the local government. The Beijing government regulated peremptory afforested areas of plants based on different industries in 1990, specifically requiring factories that produce toxic and harmful gas to accomplish afforestation no less than 40% of their land. In 2003, the China State Council centralized and reinforced administrative power for the environmental penalty, including an increased penalty for illegal storage of coal, transportation, loading, or storage of toxic and hazardous gases or dust-emitting substances. A new environmental protection law in 2018 grants the local government the right to shut down enterprises whose pollutants or emissions exceed the standard.

In recent years, however, single use of a non-market-based EP tool has proved to be unsustainable and only effective in the short term [13,14]. Countries with commitments under the Kyoto Protocol to limit or reduce greenhouse gas emissions must meet their targets primarily through national measures. As an additional means of meeting these targets, the Kyoto Protocol introduced three market-based mechanisms, thereby creating what is now known as the carbon market. These mechanisms help countries that have made Kyoto commitments meet their targets by reducing emissions or reducing carbon from the atmosphere in other countries cost-effectively. This approach encourages the private sector and emerging countries to contribute to emission reduction efforts. The Kyoto Protocol Mechanism created an international framework for market-based regulation and stimulated the development of greenhouse gas emission trading mechanisms at the national and international levels. The Chinese government has been encouraged to adopt marketbased EP instruments, and China is opening up a new era of environmental protection. In 2013, the first Chinese carbon emission trading platform was launched in Shenzhen, marking a crucial step in developing China's emission trading system. In 2018, the Chinese government canceled the pollution fee, a command-and-control EP tool, and replaced it with environmental taxation, a marked-based EP tool. The startup of China's unified carbon emissions trading market in July 2021, covering more than 4 billion tons of emissions, has made it the world's largest carbon market.

Marked-based and non-marked-based EP instruments are simultaneously used in China, which raises important research questions: Will the mixed EP instrument lead to policy conflict and crowding out? Or, will the two kinds of EP instruments complement each other and generate a friendly environment that stimulates firms' STI? Will technical executives react to governments' EP instruments [15,16]? Only some studies have addressed these research issues. The ability to answer these questions is particularly important

because it will help government more effectively apply EP instruments to improve firms' STI and benefit society in sustainable innovation. Thus, we filled this gap by first exploring the single and co-effects of market-based and non-market-based EP instruments on STI. Then, we empirically analyzed the separate and interactive effect of the two kinds of EP instruments on Chinese firms' STI. We selected environmental protection tax (EPT) and EPI to represent market-based and non-market-based EP instruments and to assess the effects of advantages and disadvantages.

This paper is structured as follows. Section 2 reviews the literature on EP instruments and firms' STI. Then, we offer two hypotheses, proposing that the direct influence of EPT and EPI on firms' STI may be positive and negative, respectively. Executives with technological backgrounds could strengthen EPT's positive effect and lessen EPI's negative impact. Moreover, mixed-use EPT and EPI instruments should improve firms' STI. Section 3 reports the samples and data collecting process and data analysis strategy. Section 4 presents the empirical results, and Section 5 provides a theoretical discussion and practical suggestions.

#### 2. Literature Review and Hypotheses

#### 2.1. The Framework of EP Instruments and Innovation

#### 2.1.1. Stepping to Embrace Market-Based EP Instruments from EPT Reform

EP instruments are designed to prevent or reduce the harmful impacts of human activities on ecosystems. Thus, the ability to encourage firm innovation is a follow-up objective of EP's primary goal [15] (i.e., reduce pollutants). The function of EP to stimulate innovation was first observed by Porter [7]. Porter and Van der Linde [8] found that market-based regulation is more flexible and better for facilitating technological innovation. Market-based EP tools focus on the results of enterprises rather than intervening in their processes or directly controlling the pollution of enterprises [17]. Therefore, they will not disrupt the market operation mechanism [18,19].

Moreover, market-based EP instruments are low-cost and easy to apply. For example, in the case of EPT, the EPT rule (i.e., more emissions, more charges; fewer emissions, fewer charges) creates room for regulators to adjust the levies in the later year according to firms' marginal costs of annual pollution. Similarly, according to their business strategies, firms can decide to manufacture less to decrease their emission or maintain their manufacturing speed at the cost of the EPT.

In addition, market-based environmental regulation policies usually have reliable consistency. For example, in the case of an EPT, the change of the levy standard needs to be decided by the Standing Committee of the Provincial People's Congress rather than the local government. Therefore, the policy is less likely to be changed by the government's preference based on the economic contribution of enterprises to the local area.

Conversely, governments' command-and-control regulations have been found to be less effective from a long-term view [20], such as EPI. Additional cost enterprises pay to reduce pollution, which creates more social benefits than economic benefits for enterprises [21]. It may also decrease R&D inventiveness for upgrading technology based on a long-term view. One primary characteristic of non-market-based EP instruments is to force enterprises to bear the cost burden of public pollution control that is not directly caused by these enterprises [22]. For example, public firms constantly invest money to support public afforestation, decrease pollution, and mandate governance wastewater abatement [23].

Before implementing the Environmental Protection Tax Law, China had a longstanding sewage charging system. The implementation of the emission charging system suffers from the disadvantages of low charging standards, many administrative interventions, insufficient rigidity of enforcement, non-standard rates, and lack of standardization, which have affected the effectiveness of its role in pollution control and emission reduction [24,25]. External factors such as administrative intervention and legal circumstances inevitably affect the emission charging system and are not mandatory, enforceable, and supervisory enough. For example, some local governments are oriented to economic growth, ignore pollution problems, and collide with enterprises [26,27] to introduce unauthorized policies and regulations, affecting sewage charges' impacts and aims. The effectiveness of the EP promotes sustainable development. In addition, due to the low standard of sewage charge, the sewage charging system has not been fully effective in reducing pollution emissions [28], and even the standard of sewage charge collection is lower than the cost of pollution treatment, which results in the phenomenon of enterprises "paying money to discharge reasonably". This kind of non-market-based EP tool may intimidate firms to control pollution, but it seems that needs-market-oriented inventiveness drives firms to conduct green technology with internal motivation.

Another non-market-based EP tool is pollution charging. A pollution charging system is an important government environmental regulation tool, first introduced in the Environmental Protection Law (Trial). The effect of this system is to internalize the externalities of environmental pollution by incorporating external environmental factors into the internal decision-making of production and operation. However, during the implementation of the emission charging system, pollution shortly dropped off and problems came forth, such as lessening standards of charging, administrative interventions, insufficient rigidity in law enforcement, and non-standard local rates. When setting standards for sewage charges, local governments consider the affordability of enterprises and the local economic situation. Most policymakers use the national bottom line as the charging rate to avoid huge threats to enterprises' operations, coupled with ineffective implementation and lack of supervision. This results in the actual payment of pollution charges being much lower than the investment cost for process upgrading and advanced technology application.

Thus, the incentive for enterprises to participate in pollution control is limited, but rather triggers enterprises to "pay the legal sewage" and causes a reluctance to carry out sustainable development. Lacking mandatory consistency, the EP tool cannot drive enterprises to do environmental production or bettering technology. Even controlling pollution and innovating technological outputs in a short time is far away from sustainable innovation. That requires long-view-based tools as a backup.

In this context, we will discuss Chinese EP makers incorporating traditional EP with fresh styles, such as the 2018 environmental tax reform. Due to several classifications of EP instruments, the market-based and non-market-based categories will be selected to discuss the EP harmonization mechanism and its impacts on firms' STI. Market-based EP currently implemented in China includes EPT, emission trading, quotas in renewable resources, environmental information disclosures, etc. Compared to EP instruments that are only partially implemented in some areas or lack statistics to track implementation, EPT is nationally implemented for five years with 2-years of national statistical observation. Thus, we choose EPT as the classic marketed-based EP tool. As for the traditional non-market-based EP instruments, EPI has been consistently implemented since its inception and is aimed at all businesses.

#### 2.1.2. Market-Based and Non-Market-Based EP Instruments' Relationships with Firms' STI

EP instruments are designed to prevent and reduce the harmful impacts of human activities on ecosystems. Thus, the ability to encourage firm innovation is a follow-up objective of EP's primary goals [15] (i.e., reduce pollutants, improve the environment, energy conservation, etc.). The function of EP to stimulate innovation was first observed by Porter [7]. Subsequently, researchers studied the relationship between EP instruments and innovation and found that market-based regulation is more flexible and better for facilitating technological innovation [8]. Afterward, supportive research argued that flexible EP works better on innovation, such as technology-based standards, mission trades, and tax [18,19]. Moreover, the empirical investigation found that a carbon tax instrument reduced pollution and stimulated technological innovation in Norway and Sweden [29].

However, the existing research mainly focused on the relationship between EP tools and promoting green technology, environmental performance, and economic performance [23,30–32]. Driven by EP, firms are incentivized to develop green technology to reduce their pollution.

Sustainable innovation asks firms to change their traditional method of innovating technologies, producing goods, or providing services to gain profits as well as long-term social and environmental benefits [6], as Boons et al. argued that green technology is only one of the approaches necessary to achieve firms' STI [33]. The most fundamental driver of sustainable innovation is an internal incentive to conduct technological innovation (CNCTST, 2016). In the long term, it is necessary for policymakers to harmonize non-market-based and market-based EP to lead enterprises to conduct STI rather than only focus on pollution date drop-off.

Specifically, under heterogeneous EP instruments, firms make different choices. Enterprises can reduce the pollutants and the concentration of emissions produced in the production process to a certain extent by purchasing and installing pollution control equipment and other end-of-pipe treatments. Enterprises can also adopt ways to strengthen the front-end control of pollutants, i.e., by investing in new green technologies, etc., and to increase energy utilization, improve production processes, and reduce the amount of pollutant output per unit [34].

In general, when faced with short-term, less rigid environmental regulatory policies with weak enforcement, firms prefer to adopt the less costly short-term response, i.e., purchasing pollution control equipment. In contrast, when faced with long-term, stable, and rigid EP instruments, firms prefer to adopt a less costly long-term approach, i.e., investing in green production technologies. Therefore, under the early sewage charging system, heavy polluters preferred to increase their environmental investments. The implementation of the environmental protection "fee to tax" policy will help promote the upgrading of the environmental protection investment structure of heavy polluters. Thus, the impact of EP tools on firms' STI is underexplored and deserves more attention.

Non-market-based EP, once strictly interpreted for instantly bettering environmental performance, will decrease innovation performance [20]. Neoclassical economists have pointed out that tight policy circumstances will lead to a higher expenditure of the enterprise's business cost and, in turn, will decrease firms' willingness to invest in R&D activities or new product exploitation when resources are limited. This will decrease not only innovation but also productivity and will further inhibit economic growth [35].

When non-market EP enhances firms' burden continuously, it will lead to firms' unstainable development. Because enterprises could face potential punishment at any time as well as subsequent discriminatory treatment from banks and other financial institutions, they lose the opportunity to seek sustainable development [36]. Banks and other financial institutions usually invest only in enterprises that comply with orders and control regulations. Because of the economic system, this negative impact of non-market EP will expand under a long-term view. Non-market-based EP may be efficient at accomplishing short-term environmental protection aims, but sustainability is a marathon, not a sprint—especially for firms' STI. Thus, we propose the following hypotheses:

H1a. Market-based EP instruments, such as EPT, will increase firms' STI.

H1b. Non-market-based EP instruments, such as EPI, will decrease firms' STI.

# 2.2. *The Harmonization Mechanism of EP Instruments and Impacts on Firms' STI* 2.2.1. Policy Mix Theory

"Policy mix" usually refers to a combination of two or more regulations by different departments or institutions targeting different objects [37]. The two categories are policy objective mixes and policy instrument mixes [15]. Recently, Rogge and Reichardt [38] studied a policy mix for sustainability transitions. They argued that policy mix is not mere tools' interaction but includes the policy strategy and the mix of interacting instruments.

Sorrell [39] used "policy mix" to study the combination of emission trading and other EP instruments (including carbon/energy taxes and non-market-based regulations on renewable energy). Braathen [40] explored the interactive effect of emission trading systems, environmentally related taxes, subsidies, "command-and-control" regulations,

and information instruments on carbon dioxide emission and economic efficiency. Some scholars analyzed non-market-based EP combinations, such as how to use the industrial sulfur dioxide removal rate, industrial wastewater compliance rate, and industrial solid waste comprehensive utilization rate to construct a performance-based environmental regulation index system [41].

Greco et al. argued that mixed policy instruments from different fields work better than a single policy instrument on eco-innovation performance rather than on general technological innovation [15]. Frequently, however, policy mix finally becomes a "policy mess" when the mix of instruments lacks coherence [39] or when they are working toward conflicting goals [4,42].

#### 2.2.2. The Harmonization of Non-Market-Based and Market-Based EP Instruments

In this study, we proposed that the mixed-use of market-based and non-marketbased EP instruments would produce harmonious results for firms' STI. According to the "narrow" vision of the Porter hypothesis, the market-based EP approach provides firms with autonomy and helps them achieve eco-friendly development [20]. The process of environmental protection, however, is filled with uncertainty and information asymmetry. Thus, it often is in opposition to rational expectations and deviates from perfect market conditions. For example, exactly how much pollution should be controlled and how to control pollution in terms of specific indicators and approaches have a large amount of uncertainty. Environmental pollutants and their consequence are unpredictable. Exploring green technology requires a huge investment with a long return cycle and also has double externality ("knowledge spillover" and "environmental spillover"). It is highly risky for firms to gain consequent benefits [43]. Therefore, it is difficult to motivate firms to only depend on market-based policy instruments to satisfy both the goals of eco-innovation and technological innovation.

Market-based EP instruments might not be good at handling time. An effective EP portfolio might include both market and non-market-based instruments. A similar opinion was offered by Howlett and Rio [44], who concluded that an EP mix could be beneficial rather than a mess. They found that both hard and soft EP instruments could complement each other and work toward achieving a common goal of encouraging the eco-innovation of firms.

In this study, we investigated a mixed usage of EPT (market-based EP) and EPI (nonmarket-based EP). Usually, firms' share of governance fees for public environmental issues equals (or is less than) their marginal social costs or damage incurred. When governments only use non-market-based EP instruments, however, firms might face the risk of having to cover more than their share. Policymakers tend to force firms to take on an exact share of pollution control through every command-and-control regulation, regardless of the firms' costs [45]. So, the superposition of non-market-based instruments likely creates overfunding and overly constrains the market due to the cost exceeding the marginal abatement for pollution.

On the other hand, a market-based EP mechanism potentially incentivizes firms to decrease their marginal abetment cost and realize the equi-marginal principle [46]. It could be argued that the practice of market-based EP instruments complemented with a command-and-control policy could be cost-effective and efficient. Specifically, EPT can be used to balance the potential negative effects of EPI on the uncertainty of firms' costs on the environment. Taxation rates and tax targets are both adjustable by the government to cover the full environmental marginal damage [47]. EPT also provides a space for firms to select the most appropriate strategy for their business. If they want to develop technology and enlarge their business, which could result in more pollution, the EPT rate will provide the best estimate of how much they should be paid for this development. After considering the development price and value, firms would be able to make the most effective decision based on technological innovation to achieve sustainable development. Hence, EPT could

be complementary and could be combined with EPI as an interlocking package. For these reasons, we proposed the following hypothesis:

**H2.** *A flexible EP portfolio, which is a combination of market and non-market-based EP instruments, will positively coinfluence firms' STI.* 

#### 2.3. The Top Manager's Reaction to EP Implementation and the Moderate Effect on Firms' STI

Firms' investment in technology innovation depends on the characteristics of their top management team (TMT). According to the upper-echelon perspective, Hambrick and Mason [48] suggested that the TMT plays a key role in offering visions, setting goals for the enterprise, and determining firms' efforts toward organizational innovation [49]. Organizational outcomes are better understood by a TMT than by a single CEO [50]. TMTs play an important role in shaping organizational characteristics by taking part in the firm's key decision-making processes [51]. The correlation is high between TMT characteristics and TMT visions and consensus toward organizational innovation, learning, or performance [52–54].

The cognitive and intellectual abilities of top executives are shaped by their work experiences, according to psychological research. Specifically, their job characteristics and experiences of success or failure shape their cognitive preferences, attitudes, and intellectual characteristics [55]. In terms of technological innovation, top executives with long-term technical experience in areas such as engineering, design, and research and development are more likely to value investment in R&D and recognize the long-term value of sustainable technological innovation for the firm. This can make them better equipped to interpret and respond to sustainable environmental policies within the organization.

Furthermore, a technical background provides the executive team with a higher level of professional experience and technical competence, which allows them to evaluate the risks of green innovation activities and be more confident in the face of green innovation uncertainty. In fact, research suggests that a technical executive team is more open to new technologies, products, and ideas and is also more tolerant of failures in sustainable technology innovation [56]. Thus, TMT, with its technical background, could balance green technology risk, R&D for technological innovation, and social responsibility to achieve sustainable progress.

In addition, based on expected utility theory, personal experience will make individuals' decision-making preferences align with their professional knowledge and thus produce behavioral biases in organizations [57]. Different experiences of corporate executives will introduce different impacts on firms' behaviors and strategies [58,59]. For example, production managers tend to pay more attention to production issues, whereas sales supervisors tend to improve the performance of firms' sales [60]. If the executives have technological experience, then their organizations likely will prefer technological investment and will emphasize R&D activities [61,62]. Executives with production, technology, or R&D experience also prefer to pay attention to and understand technological developments and prefer to invest in product and technological innovation [63]. Han et al. found that executives with technology and R&D backgrounds usually are characterized by creativity, and they play an essential role in promoting product innovation strategies and technological innovation capabilities [64]. Inventor executives with a deep understanding of the technological frontier can facilitate corporate innovation management and identify more innovation opportunities from regulations and policy [65]. Thus, technology executives may influence the company's response to EPT and EPI rather than "transferring" funds from planned R&D activities to paying for EP costs. Their decision-making process about further R&D and product plans may be less influenced by EPT than by EPI. Taken together, executives with technological backgrounds indirectly enhance firms' STI. Thus, we proposed the following hypothesis:

**H3a.** TMT with technical background will magnify the positive relationship between market-based policy instruments and firms' STI.

**H3b.** *TMT* with technical technology background will decrease the negative relationship between non-market-based policy instruments and firms' STI.

After literature review above and we developed a framework (see Figure 1.) to expand the research.



Figure 1. The research framework and hypotheses.

### 3. Methodology and Research Design

3.1. Samples and Data Collecting Process

In this paper, sample firms were A-Share Listed companies in China from 2015 to 2019. Since this paper uses the DID method to study the impact of the Chinese environmental protection tax that started to be implemented in 2018, the years 2015 to 2017 were selected as the control group in this paper. In addition, the duration was selected due to data on firms' R&D investment prior to 2015 being unavailable, and data on R&D investment later than 2019 has not been published. Then, we processed the original data as follows: (1) ST and \*ST firms were moved because their operating data were not representative; (2) firms belonged to the financial industry according to the 2012 industry classification standard of the Securities and Futures Commission were moved, because their business was special and their accounting standards were different from firms in other industries; (3) firms with missing values of R&D investment were moved; (4) firms with the value of continuous variables belong to upper 1% and lower 1% were moved. Finally, the study collected panel data from 618 firms over the period 2015–2019, resulting in a total of 3090 observations.

The main data used in the study came from three sources: One is the Chinese Research Data Service Platform (CNRD) database. It supplies the firms' R&D investments, executive information, and patent information. Data on the financial and market of the firms came from the Resset database, which includes information on firm sales growth, firm leverage, etc. Data on firms' environmental investments came from the firms' annual reports.

#### 3.2. Measuring Variables

#### 3.2.1. Dependent Variable

STI was measured by the Data Envelopment Analysis (DEA) method developed by Charnes [66]. DEA has been widely used in literature for the analysis of performance with multiple outputs and multiple inputs [67]. The DEA method includes two models: the CCR model and the BCC model. The BCC model is for measuring pure technical efficiency, so it is not appropriate to be used in this study because Chinese firms have a 'scale effect' on their

performance. Therefore, the CCR model of DEA, which is able to measure performance, was selected. The calculation formula of the CCR model is as follows:

min  $\vartheta_i$ 

$$\sum_{i=1}^{m} \lambda_j x_{ij} \le \vartheta_j x_0, \ i = 1, 2, \cdots, m$$
  
s.t.  
$$\sum_{r=1}^{s} \lambda_j y_{rj} \ge y_0, \ r = 1, 2, \cdots, s$$
  
$$j = 1, 2, \cdots, n$$
  
$$\lambda_j \ge 0$$

 $\vartheta_j$  is the efficiency; firms with solution  $\theta_j = 1$  are regarded as relatively efficient, or benchmark firms and their performances determine the efficient frontier, whereas firms for which  $\theta_i < 1$  are regarded as inefficient.

 $x_{ij} = (x_{1j}, x_{2j}, \cdots, x_{mj}) > 0$  is input.

 $y_{rj} = (y_{1j}, y_{2j}, \dots, y_{sj}) > 0$  is output.  $\lambda_j$  is the intensity factor that reflects the contribution of firm *j* in the derivation of the efficiency of another firm.

Moreover, we used the window-DEA method to reflect the time effect, in other words, to perform the horizontal comparison among decision-making units (DMUs) and the historical comparison of the same DMU. Following the previous literature, the window time in this study was a 3–4 year period [68–71].

Assuming that there are *j* DMUs, *t* time periods of data, and a time window width of *d*, each DMU has t - d + 1 time window and *d* efficiency value in the  $f(f = 1, 2, \dots, t - d + 1)$  time window. For each DMU at the  $e(e = 1, 2, \dots, d)$  time point within the  $f(f = 1, 2, \dots, t - d + 1)$  time window, the input-oriented calculation formula is shown below.

 $min\vartheta_i$ 

$$\sum_{i=1}^{d \times m} \lambda_j^{fe} x_{ij}^{fe} \le \vartheta_j^{fe} x_0^{fe}, i = 1, 2, \cdots, m$$
s.t.
$$\sum_{r=1}^{d \times s} \lambda_j^{fe} y_{rj}^{fe} \ge y_0^{fe}, r = 1, 2, \cdots, s$$

$$j = 1, 2, \cdots, n$$

$$\lambda_i \ge 0$$

 $x_0^{fe}$  means the *i*th input of the *j*th DMU at the *e*th time point within the *f*th window.  $y_0^{fe}$  means the *r*th output of the *j*th DMU at the *e*th time point within the *f*th window.  $\vartheta_j^{fe}$  means the optimal solution of the above model.

In the DEA model, multiple input and output factors were included in determining the efficiency of each DMU. We used R&D personnel (CNRD database) and R&D expenditure (CNRD database) as input variables. Patent grants (CNRD database), new product sales revenue, and green patents applications (CNRD database) are taken as output variables or the efficiency of each DMU. Because R&D inputs usually cannot bring outputs within one year, we, therefore, use patent grants rather than patent applications to reflect the time lag. Moreover, because the new product sales revenue on the firm level was not available, it was calculated by the average data of new product sales revenue (China Industry Statistical Yearbook 2015–2019) of the industry times firms' ROA [72] (Resset). Among the output

variables, the patent grants count had zero value and was not acceptable for the DEA model; thus, we replaced the zero value with a fairly small value of 0.0001 [73].

#### 3.2.2. Independent Variables

The market-based EP policy instrument was measured as the firm's EP tax fee. In China, the tax was implemented in January 2018.

The non-market EP policy instrument was measured by the firm's EP investment reported in the annual reports of the firms. We excluded all the punitive expenditures (such as sewage charges) and only included the government-mandated direct investments for environmental improvements. It was the sum of investment for afforestation, preventing pollution and wastewater abatement, etc. The two EP policy data were manually collected from the firms' annual reports.

#### 3.2.3. Moderating Variable

Technical executive. The data came from the CNRD database, and it regards technical executives as managers who are experts in technology or who have worked in the department of technology and production. If none of the executives of the firm had a technical background, it was given "0". Otherwise, it was given "1".

#### 3.2.4. Control Variables

Control variables of this study are from the Resset database: (1) firm age (measured by the difference in years between the founding year and the year 2019). It was found to negatively impact firm innovation [74,75]; (2) increased rate of main business revenue, leverage (measured by the ratio of total assets to shareholders' equity), was found to be positively related to firm innovation [76–80] and significantly influenced firm R&D investment [81–84]; (3) growth (measured by the ratio of the difference between the revenue from the main business in the current period and the revenue from the main business in the previous period to the revenue from the main business in the previous period); (4) cash (measured by the cash-to-assets ratio) was found to be positively correlated to R&D investments [85]; (5) cost management (measured by the management fee rate) was found negatively affect technical efficiency [86,87].

The means, standard deviations, and correlations of the main variables are presented in Table 1.

#### 3.3. Data Analysis Strategy

Firstly, we employed a difference-in-differences (DID) estimator to study the effect of EP on firm STI and a panel OLS estimator to study the policy mix effect, the moderating effect of technical executives on the relationships between EPI and STI as well as EPT and STI. DID is frequently used to study observational data regarding the effects of exogenous events when subjects cannot be randomly assigned to a treatment group and a control group [88]. The following is a linear ordinary least squares (OLS) specification with two-way fixed effects:

$$TIP_{i,t} = \beta_0 + \beta_1 post_t \times treatment_i + \beta_2 control_{i,t} + \beta_3 industry_i + \gamma_i + \mu_t + \varepsilon_{i,t}$$
(1)

The subscript I denotes a firm, and *t* is the year.  $post_t \times treatment_i$  is the interaction between a dummy code unit for firms (*treatment<sub>i</sub>*) that should pay the EPT and a dummy variable coded unit for the post–2018 ( $post_i$ ). *treatment<sub>i</sub>* was given "0", which means firms not paying the EPT, and was given "1", which means taxable firms.  $post_t$  was given "0" means years before 2018, and "1" means the year 2018 and subsequent years. The coefficient  $\beta_1$  represents a difference between firms both paying or not paying the EPT in the post-2018 period compared to the pre-2018 period. The *treatment<sub>i</sub>* and post\_t dummy is omitted, as it would be collinear with the firm and year fixed effects [89].  $\gamma_i$  is the firm fixed effects, and  $\mu_t$  and  $\varepsilon_{it}$  are error terms.

	Count	Mean	Sd	Max	STI	EPT	EPI	Technical Executive	Age	Growth	Manage Cost	Leverage	Cash
STI	3090	0.5807	0.1218	1	1								
EPT	515	0.4185	1.3953	12.633	0.0445	1							
EPI	3090	0.0147	0.0217	0.1146	-0.114 ***	0.200 ***	1						
Technical Executive	1482	0.5007	0.5002	1	-0.0190	0.0984	0.0495	1					
Age	3090	2.9310	0.2747	3.6636	-0.111 ***	0.102 *	0.140 ***	0.0336	1				
Growth	3090	0.1427	0.3464	2.1684	0.0646 ***	-0.0482	-0.0407 *	0.0104	-0.0841 ***	1			
Manage cost	3090	0.0951	0.1691	7.2843	0.0408 *	-0.165 ***	-0.0896 ***	-0.0236	-0.0883 ***	0.152 ***	1		
Leverage	3090	2.1943	5.2525	206.89	0.0363 *	0.0123	0.0439*	0.0542 *	0.0324	-0.0355 *	0.0275	1	
Cash	3090	0.1311	0.0953	0.4797	0.0411 *	-0.0860	-0.0656 ***	0.0429	-0.0224	0.00425	0.0185	-0.0567 **	1

Table 1. Summary	v statistics and	correlation	coefficients	for the k	ev variables.
2					2

Note: \* *p* < 0.05, \*\* *p* < 0.01, \*\*\* *p* < 0.001.

Then, panel OLS regressions were employed to study the rest of the hypotheses. We extracted firms paying EPT from the dataset to analyze the micro-effects of EP instruments. Moreover, we included the fixed-effect estimator in the regressions to control for unobservable firm factors that might influence STI. To be noted, during our observation period, Sino-US trade friction happened. These macro-shocks may have impacts on industries. It was found that multidimensional shocks in the real economy often lead to different reactions in the industry. The effect of the industry's reaction can be consistently and properly obtained in a panel data context [90]. Thus, to prevent these unobservable common shocks and their different impacts on industries, we also controlled the industry-time interactive fixed effect. This strategy has been widely used in recent literature [91–95]. In addition, we controlled clustered robust standard errors at the province level to avoid serial correlations and heteroskedasticity.

The main data analysis tools of this study are R and Stata17.

#### 4. Data Analysis

#### 4.1. Hypotheses Test

To test H1a, which states that EPT will increase STI, DID models were used. The results are presented in Table 2. They show that EPT firms have 1.4% higher STI than non-EPT firms (p < 0.05, see model 1). Then, industry dummies and a set of firm-level controls were added, and it turns out that EPT firms have 1.1% higher STI than non-EPT firms (p < 0.1, see model 2). Thus, all three models show that EPT has a statistically significant positive impact on STI. Thus, H1a is supported.

	Model 1	Model 2
post *taxtreat	0.01425 **	0.01151 *
-	0.00686	0.00671
Age		-0.28433 ***
0		0.06724
Growth		0.00001
		0.00087
Management cost		-0.02201 ***
Ū.		0.00595
Leverage		0.00084 **
-		0.00038
Cash		0.05451
		0.03696
Constant	0.57853 ***	0.1405 ***
	0.00104	0.19667
Fixed effects	Firm, Year	Firm, Year
Industry dummies	No	Yes
Observations	3090	3090
R-squared	0.43758	0.44250

Table 2. DID estimates on STI.

Notes: 1. Province clustered standard errors were presented under coefficients; 2. \*\*\* p < 0.001, \*\* p < 0.05, \* p < 0.10.

To test H1b and H2, we estimated fixed-effect OLS regressions. As shown in model 2 of Table 3, EPI has a negative impact on STI ( $\beta = -0.54$ , p < 0.1). Thus, H1b is supported. The interactive term of EPT and EPI positively affects STI ( $\beta = 3.7129$ , p < 0.1, see model 6). Thus, H2 is supported.

Finally, to test H3a and H3b, we added a moderating variable: Technical executive. The interaction terms Technical executive \* EPT and Technical executive \* EPI both statistically significantly predict STI ( $\beta = 0.04$ , p < 0.1;  $\beta = 1.1$ , p < 0.05; respectively). This indicates that a firm that has a technical executive is more likely to have better STI. Therefore, H3a and H3b are supported.

	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
EPT	0.04598 *		0.05068 *	0.02830	-0.01863	
	0.02669		0.02531	0.02489	0.06002	
EPI		-0.54097*	-1.14048 **	-1.07970 *		-0.80446
		0.27015	0.51908	0.53613		0.51000
EPT *EPI				3.71294 *		
				2.18392		
Technical executive					-0.03616	-0.02817 **
					0.02869	0.01358
Technical executive *EPT					0.04043 *	
					0.02192	
Technical executive *EPI						1.10375 **
						0.50639
Age	-0.27273	-0.39825 **	-0.36667	-0.30670	2.0778	-0.40521
Ū.	0.94956	0.18870	0.96546	0.94846	2.29738	0.34806
Growth	0.00858	-0.00315	0.00096	0.00418	0.01196	0.01067
	0.02272	0.00921	0.02504	0.02542	0.05570	0.01646
Manage cost	0.31397	-0.02308 ***	0.35422	0.35275	0.19871	-0.32959 *
-	0.24428	0.00598	0.26318	0.26317	0.85678	0.17359
Leverage	0.00065 **	0.00149 ***	0.00082 ***	0.00080 ***	0.02458	0.00168
_	0.00028	0.00031	0.00028	0.00028	0.08479	0.00133
Cash	0.05303	0.08639	0.08965	0.07935	-0.25673	0.03552
	0.10183	0.05504	0.11073	0.11180	0.22368	0.0718224
_Cons	1.29884	1.752701 ***	1.602386	1.428695	-5.802588	1.823488 *
	2.870439	0.5569115	2.918955	2.866922	7.044497	1.023708
Fixed offects	Firm, Year	Firm, Year	Firm, Year	Firm, Year	Firm, Year	Firm, Year
Fixed effects	*Industry	*Industry	*Industry	*Industry	*Industry	*Industry
Observations	515	1395	515	515	252	670
R-squared	0.7094817	0.5323189	0.7182822	0.7210598	0.9148703	0.7821413

Table 3. Fixed effects OLS regressions on STI.

Notes: 1. Province clustered standard errors were presented under coefficients; 2. \*\*\* p < 0.001, \*\* p < 0.05, \* p < 0.10.

#### 4.2. Robustness Check

Firstly, the DID approach of policy effect believes that even when there are no policy changes, trends in STI of treated and untreated firms would change over time in parallel. To test the parallel trend, this study uses an event study method and sets up the following regression model based on the calculation formula of (1)  $TIP_{i,t} = \beta_0 + \beta_1 post_t \times treatment_i + \beta_2 control_{i,t} + \beta_3 industry_i + \gamma_i + \mu_t + \varepsilon_{i,t}$ 

$$TIP_{i,t} = \alpha + \sum_{j=-3}^{2} \beta_j TAX_{i,j} + \gamma_i + \mu_t + \varepsilon_{i,t}$$
<sup>(2)</sup>

 $TAX_{i,j}$  is a dummy variable that takes the value of "1" when firm *i* paid EPT in the year *j* and "0" otherwise. Thus,  $\beta_0$  is the effect in the period of implementing EPT,  $\beta_{-3}$  through  $\beta_{-1}$  are the effects in periods 1–3 before the implementation, and  $\beta_1$  through  $\beta_2$  are the effects in periods 1–2 after the implementation. Period one before the implementation was used as the base group of the model [96] and, thus, was not included in the regression. If  $\beta_{-3}$  through  $\beta_{-2}$  are significantly zero, then the common trend assumption is established, and  $\beta_1$  through  $\beta_2$  are the effects after the implementation. The coefficients of  $\beta_{-3}$  and  $\beta_{-2}$  are not significantly different from 0, and  $\beta_0$  is significantly different (see model 1 of Table 4). Thus, the parallel trend passed the examination. In addition, we also employed a placebo test by changing the dependent variable to firm age, which is not affected by EPT (see model 2 of Table 4), and sample firms from retailing industry are not affected by EPT policy as the treatment group (see model 3 of Table 4). Both *post* \* *taxtreat* and *post* \* *faketreat* are statistically insignificant, which verified the robustness of the conclusions of H1a.

	Model 1 (STI)	Model 2 (Age)	Model 3 (STI)
before3tax	0.0054983		
	0.0074274		
before2tax	0.0160232		
	0.0095138		
currenttax	0.0541049 ***		
	0.0084794		
after1tax	-0.0123378		
	0.0101929		
$post \times taxtreat$		-0.0048147	
-		0.002947	
post $ imes$ aketreat			0.0015744
			0.0111456
Age	-0.0038325		-0.2907942 ***
	0.0081176		0.0675471
Growth	-0.0002707	0.0001709	0.0000222
	0.0005769	0.0004822	0.0008651
Manage cost	0.002272	0.0033239	-0.0219516 ***
	0.0068008	0.0038152	0.0059157
Leverage	0.0008056 ***	0.0000435	0.0008696 **
	0.0002618	0.0000629	0.0003918
Cash	-0.0011249	-0.0130806	0.0573884
	0.0178809	0.0105779	0.0366584
_cons	0.6183107 ***	2.93304 ***	1.425498 ***
	0.0202456	0.0013076	0.1976628
Industry dummies	Yes	Yes	Yes
Firm fixed	Yes	Yes	Yes
Time fixed	Yes	Yes	Yes
Observations	3090	3090	3090
R-squared	0.2479216	0.9935008	0.4420072

Table 4. DID common trend assumption and placebo tests.

Notes: 1. Province clustered standard errors are presented under coefficients; 2. \*\*\* p < 0.001, \*\* p < 0.05; 3. post × faketreat denotes the interactive term of firms belonging to retailing industry and the year of post-2018; 4. "before3tax", "before2tax", "currenttax after1tax" denote  $\beta_{-3}$ ,  $\beta_{-2}$ ,  $\beta_0$ ,  $\beta_1$ , respectively.

Secondly, to prevent the sample selection bias from coming from missing data of the dummy variable technical executive (presence of technical executives), the two-stage Heckman method is applied to identify and correct the selection bias. In the first stage, a dummy variable of whether the technical executive has missing values was used as the dependent variable in a probit regression. Covariates include total assets and the number of years of company establishment was controlled. In the second stage, the inverse Mills ratio (IMR, obtained from the first stage) was then examined as a control variable. It turns out that IMR is not significant. Thus, sample selection bias is less concerning. Table 5 shows that the significance of the interactive terms remained insignificant, which means the missing value of technical executive did not cause serious sample selection bias.

Thirdly, we used an alternative estimation model to check the robustness of the result. The efficiency value calculated by DEA is between 1 and 0, and, thus, many studies use the tobit model to deal with the data within a certain range [97]. Therefore, we estimated with random-effects tobit regression with bootstrapping methods (N = 1000) to check the robustness. It turns out that  $\beta$  is 2.1058 (p < 0.05). We also employed the method proposed by Bai [90] to estimate the panel data models with interactive fixed effects by using Sino-US trade friction as the common factor. The result indicated that  $\beta$  is 3.7282 (p < 0.10).

	Model 1	Model 2
EPT	-0.0306389	
	0.0609536	
EPI		-0.8026819
		0.51504
Technical executive	-0.03914	-0.02807 *
	0.0269629	0.0137896
Technical executive $\times$ EPT	0.0436851 **	
	0.0203887	
Technical executive $\times$ EPI		1.100706 **
		0.5144984
IMR	1.980804	-0.0344477
	2.640215	0.2528263
Age	3.076095	-0.4023683
Ũ	3.060587	0.3500913
Growth	0.0222027	0.0108425
	0.0585588	0.016819
Manage cost	-0.0378281	-0.3309729 *
0	0.9370732	0.1770222
Leverage	0.0412988	0.0016691
0	0.0833313	0.0013516
Cash	-0.2717694	0.0364164
	0.2194006	0.0723632
_cons	-10.38223	1.842441 *
	10.99439	1.042354
Fixed effects	Firm, Year $ imes$ Industry	Firm, Year $ imes$ Industry
Observations	252	670
R-squared	0.9159613	0.7821674

Table 5. Two-stage Heckman examinations.

Notes: 1. Province clustered standard errors under coefficients; 2. \*\* p < 0.05, \* p < 0.10.

Taken together, the three robustness checks all support H2.

#### 5. Conclusions

This study explores Chinese EP instrument harmonization and its impact on firms' sustainable technological innovation. We observed 618 Chinese listed firms with constant technological productivity lasting five years. According to DID model results, China's environmental tax reform in 2018 significantly incentivized firms' STI. Fixed panel OLS results showed that the harmonization of EPT and EPI is complementary and significantly promotes firms' STI jointly. In further research on the effect mechanism, we testified top management team with technical background could increase the flexibility of environmental policy implementation, thereby amplifying positive impacts and dampening negative ones. The findings of this study have theoretical contributions and are suggestive of firm innovation management strategy and government policymakers.

#### 5.1. Main Findings and Their Theoretical Contributions

Following the seminal work done by Porter and his colleagues, the impacts of EP instruments on firms' innovation have gained much research attention in recent decades at the country level [4,98,99], industry level, and firm level [100,101]. In exploring the influence of EP instruments on firms' technological innovation, the main research conclusions agree with the 'narrow' hypothesis arguing that proper and flexible EP instruments implementation could advance innovation [8]. According to their study, market-based EP should be more flexible, tending to focus on the protection outcomes to set tax rates or technology standards, compared with strict non-market-based EP requiring direct expenditures in the protection process.

Recently the discussion of mixed EP instruments has been gaining research attention, and scholars believe that the characteristic of proper and flexible EP combination includes

the coherent goals of policy instruments [102,103]. However, there is no empirical evidence of the co-effect of the market and non-market-based EP instruments on firms' STI. This study is the first time this research question has been explored. The findings of this study have the following academic contributions:

First, market-based EP instrument EPT facilitates firms' STI. The main reason is that market-based EP internalizes the environmental costs borne by society and adjusts the production and product costs of firms using price instruments [104]. It is a cost-efficiency tool to reduce pollution and avoid decreasing the incentive for R&D. For example, China EPT brings that external cost into the transaction, ensuring that the buyer pays the full marginal social cost of the good. Firms pay for their pollutants or emissions through the taxation system, not instantly paying money. This part of the business cost may not be added to the price of their products. Thus, the incentive provided by the tax ensures enterprises afford their public share of environmental protection. The consistency and certainty of taxation offer some choice space for firms' further management strategies to face different market circumstances; specifically, they could adjust this part of the cost by enlarging or restraining production in a thriving or sluggish economy to accomplish firms' sustainability.

Second, the non-market-based policy instrument EPI is negatively related to STI. This finding keeps in line with the research by Stavins [17], who found that non-market-based policy would freeze enterprises' capital which was R&D usage at the outset from a long-term view. Single use of EPI for several years may cause stringent EP circumstances, which will depress R&D activities and lead to unsustainable innovation. Strict EP circumstances put firms in the dock to make innovation strategies under a changeable market. It usually focuses on the process of environmental protection, such as EPI requiring firms to invest a certain amount of money for afforestation or river restoration. Facing downward economic pressure, firms will meet serious difficulties in accomplishing sustainable innovation because the EPI fees squeeze out other expenditures, including money for R&D activities.

Third, the two types of EP instruments create a positive impact on firms' STI. This conclusion fills the gap of mixed EP impact on technological innovation performance in a sustainable view. Scholars are starting to think much of mixed policy towards technology innovation [105]; for example, Greco et al. put an environmental policy together with an innovation policy to explore the interaction effect on technological innovation [15]. Others broadly study the influence of mixed policy on innovation [38,106,107]. Based on that, our study put forward the interaction between market and non-market-based EP instruments and found this portfolio is beneficial for advanced firms' STI.

Furthermore, recent literature suggests that firms' TMT will influence firms' technological innovation [108]. The findings are similar to the previous conclusion that TMT with a technical background promotes firms' R&D expenditures and technological innovation [109]. This study pushes it forward and demonstrates that technical executives moderated the impact of EPT and EPI on firms' STI. Thus, it is the first time demonstrating that the effect of EP instruments also depends on firms' TMT technical backgrounds, which addresses the attention of enhancing a long-term sustainable view of innovation management.

In sum, according to Boons et al., sustainable innovation includes the comprehensive picture of technological innovation rather than the firms' patent count of green technology [33]. However, a pile of papers has tested how Chinese policy instruments influence green inventions [23,30–32]. They used firms' patents of green technologies to represent the sustainable innovation of the firm. This is not quite accurate. The STI performance shall include not only green technology but also constant technological innovation, commercialization, and social benefits. This study contributes to the Porter hypothesis by demonstrating that non-market-based policy instruments of EP policy tools can complement market-based EP policy tools in facilitating firms' STI, and both kinds of EP policy instruments' impacts will be positively moderated by TMT with a technical background.

#### 5.2. Policy Implications

According to the findings of this study, we suggest the following suggestions to policymakers:

Firstly, to help firms improve sustainable technological innovation, governments could mainly depend on market-based EP instruments and achieve both economic and environmentally sustainable development.

Secondly, the government should remain cautious in applying non-market-based EP instruments [110]. The EPI in this study may increase firms' costs and restrain their motivation to follow R&D expansion. The instrument may disrupt the effective price mechanisms of the market that allows producers and consumers to respond to real market situations. Although it raises firms' attention to protect the environment, it fails to be an incentive for firms' R&D expenditure and their generation of new processes and technology.

Thirdly, mixed application of market and non-market-based EP instruments should enlarge the positive effect of market-based policy instruments on firms' STI. This gives government flexibility as well as effectiveness in regulating firms' environmental protection behaviors. A piece of similar evidence was found in the northwest region of England [111]. This study responds to the appeal of "strategic policy intelligence", which requires a dynamic view of EP instruments interaction and gives additional evidence to the "weak" version of Porter's hypotheses [37]. We agree with Tidd that the innovation issue is not the only issue of making EP, and governments should use market and non-market-based mixed EP portfolios to stimulate firms' STI [112].

Moreover, in order to positively react towards EP instruments, a firm would better involve executives with technical backgrounds. They will help firms to understand technological developments, invest in R&D [63], apply product innovation strategies [64], and identify more innovation opportunities [65]. The most important role of enhancing TMT's creativity through technical executives is aiming to better interpret EP instruments, rather than conducting innovation management.

#### 5.3. Limitation and Further Research

The limitations of this study are as follows:

First, the sample firms all came from China. To generalize the findings of this study to other countries, more evidence from other countries should be investigated.

Second, this study only used EPT to stand for market-based policy instruments and EPI to represent non-market-based policy instruments exploring the respective impacts and co-effect on firms' STI. EP instruments are quite enriched; for example, beyond the EPT and EPI, emissions trading, technology standards setting, pollution penalties, and government subsidy policy also play major roles in addressing the issue of sustainable innovation [19]. The reason our study applies EPT is that it is China's most recent policy reformation since January 2018 from a direct pollution fee which is a non-market-based EP to a marked-based EP instrument. Moreover, the emission trading system just operated in July 2021; thus, its effectiveness is still waiting to be examined. As for EPI, it has been implemented for a long time, and the data of selected samples is coordinated. In the future, more instruments of EP need to be studied to obtain a more robust conclusion on the positive effect of market-based policy instruments and mixed usage of both types of policy instruments on firms' STI.

Third, in this study, we measured TMT with technical background by collecting data on whether there is an executive with a technological background in the teams. In the future, the number of technology executives and the decision-making scope data would also need to be collected by a survey to more comprehensively reflect their technological expertise and preference for technological innovation expenditure. Since CEOs and general executives hold differing levels of decision-making power and influence within their teams, group regressions may be employed to examine potential heterogeneity in the interpretation of environmental policies between CEOs and general executives with technical backgrounds. Such analyses would shed light on how technical expertise among executives of different levels may impact their understanding and response to environmental policy. Last but not least, further analysis needs to start from the perspective of industry heterogeneity to observe and explore the mechanisms by which different environmental policies affect innovation performance in different industries. Especially for manufacturing enterprises in different technology fields, it is worthwhile to analyze in depth what combination of environmental policies can promote sustainable technological innovation more effectively while maximizing environmental benefits. We hope to make practical and meaningful policy recommendations for the development of sustainable manufacturing in China.

**Author Contributions:** Conceptualization, J.J.; Methodology, Q.Z.; Software, Q.Z. and Y.H.; Formal analysis, J.J. and Q.Z.; Resources, Q.Z.; Data curation, Q.Z. and Y.H.; Writing—original draft, J.J.; Writing—review & editing, J.J. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by [National Natural Science Foundation of China] grant number [71974178;71932009;71673264].

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** Data are contained within the article. Further requests can be made to the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Coad, A.; Segarra-Blasco, A.; Teruel, M. A Bit of Basic, a Bit of Applied? R&D Strategies and Firm Performance. J. Technol. Transf. 2020, 46, 1758–1783.
- Charter, M.; Clark, T. Sustainable Innovation: Key Conclusions from Sustainable Innovation Conferences 2003–2006 Organised by the Centre for Sustainable Design. Available online: http://cfsd.org.uk/Sustainable%20Innovation/Sustainable\_Innovation\_ report.pdf (accessed on 23 February 2023).
- 3. Charter, M.; Tischner, U. Sustainable Solutions: Developing Products and Services for the Future; Routledge: London, UK, 2017.
- Rennings, K. Redefining Innovation—Eco-Innovation Research and the Contribution from Ecological Economics. *Ecol. Econ.* 2000, 32, 319–332. [CrossRef]
- 5. Kivimaa, P.; Kern, F. Creative Destruction or Mere Niche Support? Innovation Policy Mixes for Sustainability Transitions. *Res. Policy* **2016**, *45*, 205–217. [CrossRef]
- Adams, R.; Jeanrenaud, S.; Bessant, J.; Denyer, D.; Overy, P. Sustainability-Oriented Innovation: A Systematic Review. Int. J. Manag. Rev. 2016, 18, 180–205. [CrossRef]
- 7. Porter, M. America's Green Strategy; Scientific American: New York, NY, USA, 1991.
- 8. Porter, M.E.; Linde, C. van der Toward a New Conception of the Environment-Competitiveness Relationship. *J. Econ. Perspect.* **1995**, *9*, 97–118. [CrossRef]
- Sarkodie, S.A. Failure to Control Economic Sectoral Inefficiencies through Policy Stringency Disrupts Environmental Performance. Sci. Total Environ. 2021, 772, 145603. [CrossRef]
- Chen, L.; Zhou, R.; Chang, Y.; Zhou, Y. Does Green Industrial Policy Promote the Sustainable Growth of Polluting Firms? Evidences from China. *Sci. Total Environ.* 2021, 764, 142927. [CrossRef]
- 11. Mina, D.; Hadi, S.; Jalal, A. The Incorporated Environmental Policies and Regulations into Bioenergy Supply Chain Management: A Literature Review. *Sci. Total Environ.* **2022**, *820*, 153202.
- LaCount, M.D.; Haeuber, R.A.; Macy, T.R.; Murray, B.A. Reducing Power Sector Emissions under the 1990 Clean Air Act Amendments: A Retrospective on 30 Years of Program Development and Implementation. *Atmos. Environ.* 2021, 245, 118012. [CrossRef]
- 13. Albrizio, S.; Kozluk, T.; Zipperer, V. Environmental Policies and Productivity Growth: Evidence across Industries and Firms. *J. Environ. Econ. Manag.* **2017**, *81*, 209–226. [CrossRef]
- 14. Zhang, D.; Zheng, M.; Feng, G.-F.; Chang, C.-P. Does an Environmental Policy Bring to Green Innovation in Renewable Energy? *Renew. Energy* **2022**, *195*, 1113–1124. [CrossRef]
- 15. Greco, M.; Germani, F.; Grimaldi, M.; Radicic, D. Policy Mix or Policy Mess? Effects of Cross-Instrumental Policy Mix on Eco-Innovation in German Firms. *Technovation* **2020**, *117*, 102194. [CrossRef]
- Dolfsma, W.; Seo, D. Government Policy and Technological Innovation—A Suggested Typology. *Technovation* 2013, 33, 173–179. [CrossRef]
- 17. Stavins, R.N. Experience with Market-Based Environmental Policy Instruments. In *Handbook of Environmental Economics*; Elsevier: Amsterdam, The Netherlands, 2003; Volume 1, pp. 355–435.

- Jaffe, A.B.; Palmer, K. Environmental Regulation and Innovation: A Panel Data Study. *Rev. Econ. Stat.* 1997, 79, 610–619. [CrossRef]
- 19. Tu, Z.; Shen, R. Can the Emissions Trading Mechanism Realize the Porter Effect in China? Econ. Res. 2015, 50, 160–173.
- 20. Lanoie, P.; Laurent-Lucchetti, J.; Johnstone, N.; Ambec, S. Environmental Policy, Innovation and Performance: New Insights on the Porter Hypothesis. J. Econ. Manag. Strategy 2011, 20, 803–842. [CrossRef]
- Li, G.; Wang, X.; Su, S.; Su, Y. How Green Technological Innovation Ability Influences Enterprise Competitiveness. *Technol. Soc.* 2019, 59, 101136. [CrossRef]
- 22. Helfand, G.E. Standards versus Standards: The Effects of Different Pollution Restrictions. Am. Econ. Rev. 1991, 81, 622–634.
- Wang, B.; Qi, S. The Effect of Market-Oriented and Command-and-Control Policy Tools on Emissions Reduction Innovation: An Empirical Analysis Based on China's Industrial Patents Data. *China Ind. Econ* 2016, 6, 91–108.
- 24. Wang, H.; Wheeler, D. Endogenous Enforcement and Effectiveness of China's Pollution Levy System; World Bank Publications: Washington, DC., USA, 2000; Volume 2336.
- 25. Maung, M.; Wilson, C.; Tang, X. Political Connections and Industrial Pollution: Evidence Based on State Ownership and Environmental Levies in China. *J. Bus. Ethics* **2016**, *138*, 649–659. [CrossRef]
- Wu, S.; Han, H. Energy Transition, Intensity Growth, and Policy Evolution: Evidence from Rural China. *Energy Econ.* 2022, 105, 105746. [CrossRef]
- Yan, X.; Tuo, H.; Lai, Y. A Two-Way Fixed Effects Estimation on the Impact of Industrial Land Supply on Environmental Pollution in Urban China. *Int. J. Environ. Res. Public Health* 2022, 19, 14890. [CrossRef] [PubMed]
- Tong, D.; Cheng, J.; Liu, Y.; Yu, S.; Yan, L.; Hong, C.; Qin, Y.; Zhao, H.; Zheng, Y.; Geng, G. Dynamic Projection of Anthropogenic Emissions in China: Methodology and 2015–2050 Emission Pathways under a Range of Socio-Economic, Climate Policy, and Pollution Control Scenarios. *Atmos. Chem. Phys.* 2020, 20, 5729–5757. [CrossRef]
- 29. Metcalf, G.E. Market-Based Policy Options to Control US Greenhouse Gas Emissions. J. Econ. Perspect. 2009, 23, 5–27. [CrossRef]
- Wang, F.; He, Z. Environmental Regulation, Green Innovation and Industrial Structure Upgrading. *Stat. Decis. Mak.* 2022, 38, 73–76.
- Wang, C.; Lv, M.; Qiao, P. Does International Expansion Promote Multinational Enterprises'Green Innovation?—The Moderating Effect of Environmental Regulation and Industry Regulation. *Sci. Technol. Prog. Countermeas.* 2022, 40, 81–91. Available online: http://www.kjjb.org/EN/10.6049/kjjbydc.2021090353 (accessed on 23 February 2023).
- 32. Yu, Y.; Zhang, H.; Zhang, P. The Research of the Yardstick Phenomenon and Mechanisms of Asymmetric Environmental Regulation. *Manag. World* **2021**, *37*, 134–147.
- 33. Boons, F.; Montalvo, C.; Quist, J.; Wagner, M. Sustainable Innovation, Business Models and Economic Performance: An Overview. J. Clean. Prod. 2013, 45, 1–8. [CrossRef]
- Guo, Q.; Zhou, M.; Liu, N.; Wang, Y. Spatial Effects of Environmental Regulation and Green Credits on Green Technology Innovation under Low-Carbon Economy Background Conditions. *Int. J. Environ. Res. Public Health* 2019, 16, 3027. [CrossRef]
- 35. Jaffe, A.B.; Stavins, R.N. Dynamic Incentives of Environmental Regulations: The Effects of Alternative Policy Instruments on Technology Diffusion. *J. Environ. Econ. Manag.* **1995**, *29*, S43–S63. [CrossRef]
- 36. Sahoo, P.; Nayak, B.P. Green Banking in India. Indian Econ. J. 2007, 55, 82–98. [CrossRef]
- 37. Flanagan, K.; Uyarra, E.; Laranja, M. Reconceptualizing the 'Policy Mix' for Innovation. Res. Policy 2011, 40, 702–713. [CrossRef]
- 38. Reichardt, K.; Rogge, K. How the Policy Mix Impacts Innovation: Findings from Company Case Studies on Offshore Wind in Germany. *Environ. Innov. Soc. Transit.* 2016, *18*, 62–81. [CrossRef]
- 39. Sorrell, S.; Sijm, J. Carbon Trading in the Policy Mix. Oxf. Rev. Econ. Policy 2003, 19, 420–437. [CrossRef]
- Braathen, N.A. Interactions Between Emission Trading Systems and Other Overlapping Policy Instruments; OECD Publishing: Paris, France, 2011.
- 41. Wu, H.; Xu, L.; Ren, S.; Hao, Y.; Yan, G. How Do Energy Consumption and Environmental Regulation Affect Carbon Emissions in China? New Evidence from a Dynamic Threshold Panel Model. *Resour. Policy* **2020**, *67*, 101678. [CrossRef]
- 42. Beise, M.; Rennings, K. Lead Markets and Regulation: A Framework for Analyzing the International Diffusion of Environmental Innovations. *Ecol. Econ.* **2005**, *52*, 5–17. [CrossRef]
- 43. Cai, W.; Li, G. The Drivers of Eco-Innovation and Its Impact on Performance: Evidence from China. J. Clean. Prod. 2018, 176, 110–118. [CrossRef]
- 44. Howlett, M.; Del Rio, P. The Parameters of Policy Portfolios: Verticality and Horizontality in Design Spaces and Their Consequences for Policy Mix Formulation. *Environ. Plan. C Gov. Policy* **2015**, *33*, 1233–1245. [CrossRef]
- 45. Joskow, P.L.; Schmalensee, R.; Bailey, E.M. The Market for Sulfur Dioxide Emissions. Am. Econ. Rev. 1998, 88, 669–685.
- Newell, R.G.; Stavins, R.N. Cost Heterogeneity and the Potential Savings from Market-Based Policies. J. Regul. Econ. 2003, 23, 43–59. [CrossRef]
- 47. Goulder, L.H. Effects of Carbon Taxes in an Economy with Prior Tax Distortions: An Intertemporal General Equilibrium Analysis. *J. Environ. Econ. Manag.* **1995**, *29*, 271–297. [CrossRef]
- Hambrick, D.C.; Mason, P.A. Upper Echelons: The Organization as a Reflection of Its Top Managers. *Acad. Manag. Rev.* 1984, 9, 193–206. [CrossRef]
- 49. Yoon, S.N.; Lee, D.; Schniederjans, M. Effects of Innovation Leadership and Supply Chain Innovation on Supply Chain Efficiency: Focusing on Hospital Size. *Technol. Forecast. Soc. Change* **2016**, *113*, 412–421. [CrossRef]

- 50. Hambrick, D.C. What If the Academy Actually Mattered? Acad. Manag. Rev. 1994, 19, 11–16. [CrossRef]
- 51. Hambrick, D.C.; Cho, T.S.; Chen, M.-J. The Influence of Top Management Team Heterogeneity on Firms' Competitive Moves. *Adm. Sci. Q.* **1996**, *41*, 659–684. [CrossRef]
- 52. Ahlstrom, D.; Yang, X.; Wang, L.; Wu, C. A Global Perspective of Entrepreneurship and Innovation in China. *Multinatl. Bus. Rev.* **2018**, *26*, 302–318. [CrossRef]
- 53. Heyden, M.L.; Van Doorn, S.; Reimer, M.; Van Den Bosch, F.A.; Volberda, H.W. Perceived Environmental Dynamism, Relative Competitive Performance, and Top Management Team Heterogeneity: Examining Correlates of Upper Echelons' Advice-Seeking. *Organ. Stud.* **2013**, *34*, 1327–1356. [CrossRef]
- 54. Mihalache, O.R.; Jansen, J.J.; Van Den Bosch, F.A.; Volberda, H.W. Offshoring and Firm Innovation: The Moderating Role of Top Management Team Attributes. *Strateg. Manag. J.* **2012**, *33*, 1480–1498. [CrossRef]
- 55. Brousseau, K.R.; Prince, J.B. Job-Person Dynamics: An Extension of Longitudinal Research. J. Appl. Psychol. 1981, 66, 59. [CrossRef]
- 56. Farag, H.; Mallin, C. The Influence of CEO Demographic Characteristics on Corporate Risk-Taking: Evidence from Chinese IPOs. *Eur. J. Financ.* **2018**, *24*, 1528–1551. [CrossRef]
- Hertwig, R.; Barron, G.; Weber, E.U.; Erev, I. Decisions from Experience and the Effect of Rare Events in Risky Choice. *Psychol. Sci.* 2004, 15, 534–539. [CrossRef] [PubMed]
- 58. Yu, H.; Solvang, W.D. Incorporating Flexible Capacity in the Planning of a Multi-Product Multi-Echelon Sustainable Reverse Logistics Network under Uncertainty. *J. Clean. Prod.* **2018**, *198*, 285–303. [CrossRef]
- 59. Hambrick, D.C. Upper Echelons Theory: An Update. Acad. Manag. Rev. 2007, 32, 334–343. [CrossRef]
- 60. Dearborn, D.C.; Simon, H.A. Selective Perception: A Note on the Departmental Identifications of Executives. *Sociometry* **1958**, *21*, 140–144. [CrossRef]
- 61. Malmendier, U.; Tate, G.; Yan, J. Overconfidence and Early-Life Experiences: The Effect of Managerial Traits on Corporate Financial Policies. *J. Financ.* 2011, *66*, 1687–1733. [CrossRef]
- 62. Wang, C.-C.; Sung, H.-Y.; Chen, D.-Z.; Huang, M.-H. Strong Ties and Weak Ties of the Knowledge Spillover Network in the Semiconductor Industry. *Technol. Forecast. Soc. Change* **2017**, *118*, 114–127. [CrossRef]
- 63. Finkelstein, S. Power in Top Management Teams: Dimensions, Measurement, and Validation. *Acad. Manag. J.* **1992**, *35*, 505–538. [CrossRef]
- 64. Han, H. The Norm Activation Model and Theory-Broadening: Individuals' Decision-Making on Environmentally-Responsible Convention Attendance. J. Environ. Psychol. 2014, 40, 462–471. [CrossRef]
- 65. Francis, B.; Hasan, I.; Wu, Q. Professors in the Boardroom and Their Impact on Corporate Governance and Firm Performance. *Financ. Manag.* **2015**, *44*, 547–581. [CrossRef]
- 66. Charnes, A.; Cooper, W.W.; Rhodes, E. Measuring the Efficiency of Decision Making Units. *Eur. J. Oper. Res.* **1978**, *2*, 429–444. [CrossRef]
- 67. Bulak, M.E.; Turkyilmaz, A. Performance Assessment of Manufacturing SMEs: A Frontier Approach. *Ind. Manag. Data Syst.* 2014, 114, 797–816. [CrossRef]
- 68. Cullinane, K.; Song, D.-W.; Ji, P.; Wang, T.-F. An Application of DEA Windows Analysis to Container Port Production Efficiency. *Rev. Netw. Econ.* **2004**, *23*, 669–685. [CrossRef]
- 69. Fadzlan, S.; Muhd-Zulkhibri, A.M. Singapore Banking Efficiency and Its Relation to Stock Returns: A DEA Window Analysis Approach. *Int. J. Bus. Stud. A Publ. Fac. Bus. Adm. Ed. Cowan Univ.* 2007, 15, 83–106.
- 70. Hemmasi, A.; Talaeipour, M.; Khademi-Eslam, H.; Pourmousa, S. Using DEA Window Analysis for Performance Evaluation of Iranian Wood Panels Industry. *Afr. J. Agric. Res.* **2011**, *6*, 1802–1806.
- Pulina, M.; Detotto, C.; Paba, A. An Investigation into the Relationship between Size and Efficiency of the Italian Hospitality Sector: A Window DEA Approach. *Eur. J. Oper. Res.* 2010, 204, 613–620. [CrossRef]
- 72. Sofka, W.; Grimpe, C. Specialized Search and Innovation Performance Evidence across Europe. *RD Manag.* **2010**, *40*, 310–323. [CrossRef]
- 73. Bowlin, W.F. Measuring Performance: An Introduction to Data Envelopment Analysis (DEA). J. Cost Anal. 1998, 15, 3–27. [CrossRef]
- 74. Balasubramanian, N.; Lee, J. Firm Age and Innovation. Ind. Corp. Change 2008, 17, 1019–1047. [CrossRef]
- 75. Huergo, E.; Jaumandreu, J. How Does Probability of Innovation Change with Firm Age? *Small Bus. Econ.* **2004**, *22*, 193–207. [CrossRef]
- 76. Abernathy, W.J.; Clark, K.B. Innovation: Mapping the Winds of Creative Destruction. Res. Policy 1985, 14, 3–22. [CrossRef]
- 77. Choi, M.; Lee, C.-Y. Technological Diversification and R&D Productivity: The Moderating Effects of Knowledge Spillovers and Core-Technology Competence. *Technovation* **2021**, *104*, 102249. [CrossRef]
- 78. Evenson, R.E. Patents, R&D, and Invention Potential: International Evidence. Am. Econ. Rev. 1993, 83, 463–468.
- 79. Kortum, S. Equilibrium R&d and the Patent–R&d Ratio: Us Evidence. *Am. Econ. Rev.* **1993**, *83*, 450–457.
- Tushman, M.L.; Anderson, P. Technological Discontinuities and Organizational Environments. *Adm. Sci. Q.* 1986, 31, 439–465. [CrossRef]
- 81. Bhagat, S.; Welch, I. Corporate Research & Development Investments International Comparisons. J. Account. Econ. 1995, 19, 443–470.

- 82. Hsu, H. The Moderating Effects of Leverage and Ownership Structure on Firm Performance. S. E. Asia J. Contemp. Bus. Econ. Law 2013, 2, 73–76.
- 83. Nanda, R.; Rhodes-Kropf, M. Financing Risk and Innovation. Manag. Sci. 2017, 63, 901–918. [CrossRef]
- 84. Singh, M.; Faircloth, S. The Impact of Corporate Debt on Long Term Investment and Firm Performance. *Appl. Econ.* **2005**, *37*, 875–883. [CrossRef]
- 85. Himmelberg, C.P.; Petersen, B.C. R & D and Internal Finance: A Panel Study of Small Firms in High-Tech Industries. *Rev. Econ. Stat.* **1994**, *76*, 38–51.
- Miao, J.Y.; Cai, C.W. The Study on the Technical Efficiency of the Listed Coal Companies. In *Advanced Materials Research*; Trans Tech Publications Ltd.: Bäch, Switzerland, 2013; Volume 616, pp. 528–532.
- Nie, Y.; Nie, H. The Impact of Ownership Structure on Innovation Level: Moderating Effect Based on the Life Cycle. *BCPBM* 2021, 15, 202–210. [CrossRef]
- 88. Meyer, B.D. Natural and Quasi-Experiments in Economics. J. Bus. Econ. Stat. 1995, 13, 151–161.
- 89. Berger, A.; Roman, R. *TARP and Other Bank Bailouts and Bail-Ins around the World: Connecting Wall Street, Main Street, and the Financial System*; Academic Press: Cambridge, MA, USA, 2020.
- 90. Bai, J. Panel Data Models with Interactive Fixed Effects. *Econometrica* 2009, 77, 1229–1279.
- 91. Eberhardt, M.; Helmers, C.; Strauss, H. Do Spillovers Matter When Estimating Private Returns to R&D? *Rev. Econ. Stat.* 2013, *95*, 436–448.
- 92. Gobillon, L.; Magnac, T. Regional Policy Evaluation: Interactive Fixed Effects and Synthetic Controls. *Rev. Econ. Stat.* 2016, *98*, 535–551. [CrossRef]
- 93. Hagedorn, M.; Karahan, F.; Manovskii, I.; Mitman, K. Unemployment Benefits and Unemployment in the Great Recession: The Role of Macro Effects; National Bureau of Economic Research: Cambridge, MA, USA, 2013.
- 94. Hsiao, C.; Xie, Y.; Zhou, Q. Factor Dimension Determination for Panel Interactive Effects Models: An Orthogonal Projection Approach. *Comput. Stat.* 2021, *36*, 1481–1497. [CrossRef]
- 95. Petrova, Y.; Westerlund, J. Fixed Effects Demeaning in the Presence of Interactive Effects in Treatment Effects Regressions and Elsewhere. *J. Appl. Econ.* **2020**, *35*, 960–964. [CrossRef]
- 96. Wang, J. The Economic Impact of Special Economic Zones: Evidence from Chinese Municipalities. J. Dev. Econ. 2013, 101, 133–147. [CrossRef]
- 97. Guan, J.; Chen, K. Modeling the Relative Efficiency of National Innovation Systems. Res. Policy 2012, 41, 102–115. [CrossRef]
- Acemoglu, D.; Aghion, P.; Bursztyn, L.; Hemous, D. The Environment and Directed Technical Change. Am. Econ. Rev. 2012, 102, 131–166. [CrossRef]
- 99. Milliman, S.R.; Prince, R. Firm Incentives to Promote Technological Change in Pollution Control. J. Environ. Econ. Manag. 1989, 17, 247–265. [CrossRef]
- 100. Del Río, P.; Carrillo-Hermosilla, J.; Könnölä, T. Policy Strategies to Promote Eco-Innovation: An Integrated Framework. *J. Ind. Ecol.* **2010**, *14*, 541–557. [CrossRef]
- 101. Kerr, S.; Newell, R.G. Policy-Induced Technology Adoption: Evidence from the US Lead Phasedown. J. Ind. Econ. 2003, 51, 317–343. [CrossRef]
- Crespi, F.; Ghisetti, C.; Quatraro, F. Environmental and Innovation Policies for the Evolution of Green Technologies: A Survey and a Test. *Eurasian Bus. Rev.* 2015, *5*, 343–370. [CrossRef]
- Kemp, R.; Pontoglio, S. The Innovation Effects of Environmental Policy Instruments—A Typical Case of the Blind Men and the Elephant? *Ecol. Econ.* 2011, 72, 28–36. [CrossRef]
- 104. Bovenberg, A.L.; De Mooij, R.A. Environmental Tax Reform and Endogenous Growth. J. Public Econ. 1997, 63, 207–237. [CrossRef]
- 105. Braathen, N.A. Instrument Mixes for Environmental Policy: How Many Stones Should Be Used to Kill a Bird? *Int. Rev. Environ. Resour. Econ.* **2007**, *1*, 185–236. [CrossRef]
- 106. Meissner, D.; Kergroach, S. Innovation Policy Mix: Mapping and Measurement. J. Technol. Transf. 2021, 46, 197–222. [CrossRef]
- Magro, E.; Wilson, J.R. Policy-Mix Evaluation: Governance Challenges from New Place-Based Innovation Policies. *Res. Policy* 2019, 48, 103612. [CrossRef]
- 108. Prasad, B.; Martens, R. Top Management Team Advice-Seeking and Environmental Competitiveness Impacts on Technological Innovation. *Int. J. Technol. Manag.* 2015, *69*, 77–92. [CrossRef]
- 109. Yafei, L.; Fengrong, W.; Anran, L. Do Technological Entrepreneurs Promote Enterprise Innovation? Sci. Res. Manag. 2022, 43, 186.
- 110. Yu, X.; Wang, P. Economic Effects Analysis of Environmental Regulation Policy in the Process of Industrial Structure Upgrading: Evidence from Chinese Provincial Panel Data. *Sci. Total Environ.* **2021**, *753*, 142004. [CrossRef] [PubMed]
- 111. Uyarra, E.; Flanagan, K. From Regional Systems of Innovation to Regions as Innovation Policy Spaces. *Environ. Plan. C: Gov. Policy* **2010**, *28*, 681–695. [CrossRef]
- Tidd, J. Innovation Management in Context: Environment, Organization and Performance. Int. J. Manag. Rev. 2001, 3, 169–183.
   [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.